
diff -r 204564b899a0 -r fd43a9cd9c07 handouts/ho03.tex
--- a/handouts/ho03.tex	Mon Oct 06 02:42:49 2014 +0100
+++ b/handouts/ho03.tex	Mon Oct 06 02:43:49 2014 +0100
@@ -139,8 +139,8 @@
 \end{center} 
 
 \noindent This function copies data from a source \pcode{src}
-to a destination \pcode{dst}. It copies the data until it 
-reaches a zero-byte (\code{"\\0"}). 
+to a destination \pcode{dst}. The important point is that it
+copies the data until it reaches a zero-byte (\code{"\\0"}). 
 
 \bigskip\bigskip
 \subsubsection*{A Crash-Course on GDB}




